Author's response to reviews

Title: Granulomatus cheilitis in a patient with Crohn's disease: a case report

Authors:

John K Triantafillidis (jkt@panafonet.gr)
Flora Zervou - Valvi (jkt@panafonet.gr)
Emmanouel Merikas (jkt@panafonet.gr)
George Peros (jkt@panafonet.gr)
Ourania Nicolatou - Galitis (jkt@panafonet.gr)
Aristofanis Gikas (argikas@internet.gr)

Version: 2 Date: 9 May 2007

Author's response to reviews:

We are resubmitting the manuscript after the requested below changes:

Remove the 'Case Report' subheading
End the title ': a case report.'
Rename the 'Summary', 'Abstract'
Reduce the reference list as it is journal policy to allow a maximum of 10 references
Remove four figures, as each manuscript must contain no more than 3 figure.